Author's response to reviews

Title: An adult patient presenting with appendix perforation in appendix duplication: a case report

Authors:

   Emel Canbay (dremelcanbay@gmail.com)
   Emel Akman (emelakman1@hotmail.com)

Version: 10 Date: 17 November 2010

Author's response to reviews: see over
MS: 1454109964365803
Case report
Appendix duplication in an adult patient with appendix perforation A case report
Emel Canbay and Emel Akman
submitted to Journal of Medical Case Reports

November 17, 2010

Dear Mr Ed Bautista,

Thank you very much for your letter. We have completed last checking before publication. We have not changed anything. I confirm that this the last version before going to publish. Thank you for your efforts and helps.
We are looking forward to hearing from you soon.

Sincerely yours

Emel Canbay MD, PhD
Corresponding author